Author's response to reviews

Title: Ethics, economics and the regulation and adoption of new medical devices: case studies in pelvic floor surgery

Authors:

Sue Ross (sue.ross@albertahealthservices.ca)
Charles Weijer (cweijer@uwo.ca)
Amitram Gafni (gafni@mcmaster.ca)
Ariel Ducey (aducey@ucalgary.ca)
Carmen Thompson (ckthomps@ucalgary.ca)
Rene Lafreniere (rene.lafreniere@albertahealthservices.ca)

Version: 3 Date: 20 August 2010

Author's response to reviews:

Dear BMC Production Team

Thanks so much for the speedy processing of our paper! We have made the following changes as requested:

1. “Reference 2: please provide a functioning web link to the website.”
   We do not understand this comment - we verified the website quoted in our paper this morning, and it is functioning (at 09:30 Mountain Standard Time). The website is as follows:

2. All highlighting/tracking has been removed.
3. Authors’ contributions: these have been formatted into a paragraph.
4. Tables: the visible vertical lines have been removed.
5. We have checked for any typographical errors.

Please let me know if there are any other comments.

Sue Ross